Author's response to reviews

Title: An interprofessional approach to shared decision making: an exploratory case study with family caregivers of one IP home care team

Authors:

France Légaré (france.legare@mfa.ulaval.ca)
Dawn Stacey (Dawn.Stacey@uOttawa.ca)
Nathalie Brière (nathalie.briere@csssvc.qc.ca)
Hubert Robitaille (hubert.robitaille@crsfa.ulaval.ca)
Marie-Claude Lord (mary_11@hotmail.com)
Sophie Desroches (Sophie.Desroches@fsaa.ulaval.ca)
Renée Drolet (renee.drolet@crchudequebec.ulaval.ca)

Version: 2
Date: 26 May 2014

Author's response to reviews:

26 of May 2014

Danica Stephanie Lorrein Jose
Journal Editorial Office
BioMed Central

Object: MS: 7031576131222706 - Caregivers' perceptions of sharing decisions about location of care with an interprofessional home care team: a qualitative study

Dear Mrs Danica

We thank the reviewers for their constructive comments. We have completely revised our manuscript. In line with the suggestions of the reviewers, we have modified its title which is now: An interprofessional approach to shared decision making: an exploratory case study with family caregivers of one IP home care team. Please find our detailed responses to the reviewers in the attached document.

Sincerely yours,

France Légaré, B. Arch, MD, PhD, CCMF, FCMF
Canada Research Chair in Implementation of Shared Decision Making in Primary Care
Department of Family Medicine and Emergency Medicine
Faculty of Medicine
Laval University (Québec)
Canada G1K 7P4
Phone: (418) 525 4437
Fax: (418) 525 4194